# Co-trimoxazole to reduce mortality, transplant, or unplanned hospitalisation in people with moderate to very severe idiopathic pulmonary fibrosis: the EME-TIPAC RCT

Andrew M Wilson,<sup>1\*</sup> Allan B Clark,<sup>1</sup>
Anthony Cahn,<sup>2</sup> Edwin R Chilvers,<sup>3</sup> William Fraser,<sup>1</sup>
Matthew Hammond,<sup>4</sup> David M Livermore,<sup>1</sup>
Toby M Maher,<sup>5</sup> Helen Parfrey,<sup>6</sup> Ann Marie Swart,<sup>4</sup>
Susan Stirling,<sup>4</sup> David Thickett<sup>7</sup> and Moira Whyte<sup>8</sup>

<sup>1</sup>Norwich Medical School, University of East Anglia, Norwich, UK

Declared competing interests of authors: David Livermore reports personal fees from Accelerate (Tucson, AZ, USA), Allecra Therapeutics GmbH (Basel, Switzerland), Antabio (Labège, France), Centauri (Sandwich, UK), Entasis (Waltham, MA, USA), Integra-Holdings (Jerusalem, Israel), Meiji Holdings Co., Ltd (Yokohama, Japan), Melinta Therapeutics (Morristown, NJ, USA), Menarini Group (Florence, Italy), Mutabilis (Romainville, France), Nordic (Reading, UK), ParaPharm Development Ltd (Reading, UK), Pfizer Inc. (New York, NY, USA), Qpex Biopharma (San Diego, CA, USA), Roche Holding AG (Basel, Switzerland; Welwyn Garden City, UK), Shionogi B.V. (Amsterdam, the Netherlands), T.A.Z. (Grand Cayman, Cayman Islands), Tetraphase Pharmaceuticals, Inc. (Watertown, MA, USA), Venatorx Pharmaceuticals (Malvern, PA, USA), Wockhardt (Aurangabad, India), Zambon Pharma (Milan, Italy), Astellas Pharma (Tokyo, Japan), bioMérieux SA (Marcy l'Etoile, France), Beckman Coulter Inc. (West Sacramento, CA, USA), Cardiome (Vancouver, BC, Canada), Cepheid (Sunnyvale, CA, USA) and Merck/MSD (Rahway, NJ, USA); other fees from Dechra Pharmaceuticals (Northwich, UK), GlaxoSmithKline plc (Brentford, UK), Merck (Rahway, NJ, USA), PerkinElmer® Inc. (Waltham, MA, USA), Pfizer Inc., Roche Holding AG (Welwyn Garden City, UK), Boehringer Ingelheim (Brackness, UK), Zambon, GlaxoSmithKline Research and Development (R&D) (Brentford, UK) and Pfizer Inc. (Sandwich, UK), outside the submitted work; and was a member of the National Institute for Health Research (NIHR) Health Technology Assessment (HTA) Antimicrobial Resistance Themed Call Board (2013–14). Toby Maher reports grants and personal fees from GlaxoSmithKline R&D and AstraZeneca plc (Cambridge, UK); grants, personal fees and non-financial support from UCB Pharma (Slough, UK); personal fees from Boehringer Ingelheim, Roche Holding AG,

<sup>&</sup>lt;sup>2</sup>Department of Respiratory Medicine, Bedford Hospitals NHS Trust, Bedford, UK

<sup>&</sup>lt;sup>3</sup>ICTEM building, Hammersmith Campus, Imperial College London, London, UK

<sup>&</sup>lt;sup>4</sup>Norwich Clinical Trials Unit, University of East Anglia, Norwich, UK

<sup>&</sup>lt;sup>5</sup>Department of Respiratory Medicine, Royal Brompton & Harefield NHS Foundation Trust, Royal Brompton Hospital, London, UK

<sup>&</sup>lt;sup>6</sup>Department of Respiratory Medicine, Royal Papworth Hospital NHS Foundation Trust, Cambridge Biomedical Campus, Cambridge, UK

<sup>&</sup>lt;sup>7</sup>Birmingham Acute Care Research Group, University of Birmingham, Birmingham, UK <sup>8</sup>Medical Research Council Centre for Inflammation Research, University of Edinburgh, Edinburgh, UK

<sup>\*</sup>Corresponding author a.m.wilson@uea.ac.uk

Bayer HealthCare (Leverkeusen, Germany), Prometic (Toronto, ON, Canada), Samumed (San Diego, CA, USA), Galapagos (Mechelen, Belgium), Celgene (Summit, NJ, USA), Indalo (St Louis, KY, USA), Pliant (San Francisco, CA, USA), Blade Therapeutics, Inc. (San Francisco, CA, USA), Respivant (San Diego, CA, USA), Novartis Pharmaceuticals (Basel, Switzerland) and Bristol-Myers Squibb (New York, NY, USA); and stock options from Apellis Pharmaceuticals (Waltham, MA, USA), outside the submitted work. Helen Parfrey reports educational grants, conference attendance and personal fees from Roche Holding AG and Boehringer Ingelheim. She is a trustee of the charity Action for Pulmonary Fibrosis (Peterborough, UK). Anne Marie Swart was part of the NIHR HTA Efficient Study Designs Board 2014–16 and is currently part of the Clinical Trials Unit Standing Advisory Committee. Moira Whyte was part of the NIHR Efficacy and Mechanism Evaluation Funding Committee 2008–14. Andrew Wilson reports investigator-initiated research grant funding from Roche Holding AG.

Published July 2021 DOI: 10.3310/eme08090

# Plain English summary

### The EME-TIPAC RCT

Efficacy and Mechanism Evaluation 2021; Vol. 8: No. 9

DOI: 10.3310/eme08090

NIHR Journals Library www.journalslibrary.nihr.ac.uk

# **Plain English summary**

diopathic pulmonary fibrosis is an incurable, lung-scarring disease that causes people to feel progressively more breathless over time and to cough. It is a usually fatal disease. On average, people survive for 3–4 years after diagnosis.

Idiopathic pulmonary fibrosis may be made worse by lung infections. A previous study suggested that an antibiotic called co-trimoxazole (SEPTRIN®; Essential Generics Ltd, Egham, UK; Chemidex Generics Ltd, Egham, UK) may improve survival by reducing the frequency and severity of lung infections. Although co-trimoxazole has been used for many years, we needed to undertake a larger study before we could recommend this form of treatment.

This study tested whether or not co-trimoxazole reduces the chances of dying or being admitted to hospital among people with idiopathic pulmonary fibrosis. We enrolled 342 people to take part in the study from nearly 40 hospitals throughout the UK. Patients took either co-trimoxazole or a dummy tablet for up to 3.5 years. As well as counting the number of deaths and admissions to hospital, we measured lung function using breathing tests and patient-completed questionnaires.

Our results suggest that co-trimoxazole did not reduce the chances of dying or being admitted to hospital (for any reason or because of chest infections and/or worsening of idiopathic pulmonary fibrosis). However, we did find that the people taking co-trimoxazole coughed less than those who were not taking co-trimoxazole and their cough was less troublesome. Co-trimoxazole did not improve breathlessness score or breathing test results.

Our results do not suggest that people with idiopathic pulmonary fibrosis should take daily co-trimoxazole to prevent progression of their condition. They should, however, take antibiotics prescribed by their doctor for chest infections or for other reasons. We need to undertake other studies to specifically look at cough symptoms before we can be sure whether or not co-trimoxazole is useful at improving this symptom. Other studies using other antibiotics may be useful.

## **Efficacy and Mechanism Evaluation**

ISSN 2050-4365 (Print)

ISSN 2050-4373 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full EME archive is freely available to view online at www.journalslibrary.nihr.ac.uk/eme. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

#### Criteria for inclusion in the Efficacy and Mechanism Evaluation journal

Reports are published in *Efficacy and Mechanism Evaluation* (EME) if (1) they have resulted from work for the EME programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

#### **EME** programme

The Efficacy and Mechanism Evaluation (EME) programme funds ambitious studies evaluating interventions that have the potential to make a step-change in the promotion of health, treatment of disease and improvement of rehabilitation or long-term care. Within these studies, EME supports research to improve the understanding of the mechanisms of both diseases and treatments.

The programme supports translational research into a wide range of new or repurposed interventions. These may include diagnostic or prognostic tests and decision-making tools, therapeutics or psychological treatments, medical devices, and public health initiatives delivered in the NHS.

The EME programme supports clinical trials and studies with other robust designs, which test the efficacy of interventions, and which may use clinical or well-validated surrogate outcomes. It only supports studies in man and where there is adequate proof of concept. The programme encourages hypothesis-driven mechanistic studies, integrated within the efficacy study, that explore the mechanisms of action of the intervention or the disease, the cause of differing responses, or improve the understanding of adverse effects. It funds similar mechanistic studies linked to studies funded by any NIHR programme.

The EME programme is funded by the Medical Research Council (MRC) and the National Institute for Health Research (NIHR), with contributions from the Chief Scientist Office (CSO) in Scotland and National Institute for Social Care and Health Research (NISCHR) in Wales and the Health and Social Care Research and Development (HSC R&D), Public Health Agency in Northern Ireland.

#### This report

The research reported in this issue of the journal was funded by the EME programme as project number 12/206/09. The contractual start date was in December 2014. The final report began editorial review in January 2020 and was accepted for publication in October 2020. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The EME editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research. The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the MRC, NETSCC, the EME programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the EME programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2021. This work was produced by Wilson et al. under the terms of a commissioning contract issued by the Secretary of State for Health. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

## NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

## **NIHR Journals Library Editors**

**Professor John Powell** Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

**Professor Andrée Le May** Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Scientific Adviser (Evidence Use), Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk